Author's response to reviews

Title: Relationship of psychological and oral health statuses with self-perceived halitosis in a Jordanian population: a cross-sectional study

Authors:

Firas Q Alzoubi (firasz@just.edu.jo)
Jumana A Karasneh (jumana2003@yahoo.com)
Nidal M Daamseh (dr.ndaamsa@hotmail.com)

Version: 6
Date: 17 July 2015

Author's response to reviews: see over
Dear Dr Aida,

Thank you very much for considering our manuscript entitled “Relationship of psychological and oral health statuses with self-perceived halitosis in a Jordanian population: a cross-sectional study”, for the publication in BioMed Central Oral Health.

We were pleased with the comment of the reviewers and with the initial acceptance and made modifications to the manuscript accordingly. Please note the point-by-point response below.

Sincerely,

Firas Alzoubi, MD, FRCS
Chairman, Special Surgery Department
Consultant and Associate Professor of Otolaryngology
Faculty of Medicine
Jordan University of Science and Technology
PO Box 3030, 22110
Irbid Jordan
Email: firasz@just.edu.jo
Mobile: +962795411417
Editorial comments:

1) **Grammar:** Please go through and copy-edit the manuscript to make sure there is no grammar and spelling mistakes. The manuscript has been edited to ensure that the language is clear and free of errors. The edit was performed by professional native speaking editors.

2) **Tables:** Please include Tables 1 and 3 in the main text (like you have done for Tables 2 and 4), instead of as additional files. Please make sure Tables are in the right order.

Done.